Cottam Parkway Railway Station

Environmental Statement Volume 1: Non-Technical Summary Document Reference: ES-01-NTS-02



lancashire.gov.uk

Introduction

Lancashire County Council is applying for planning permission for a new railway station to provide transport infrastructure to North West Preston.

As part of the planning process, an Environment Impact Assessment (EIA) has been carried out to understand the potential effects that the new railway station would have on the environment. An EIA investigates the environmental consequences of a plan, policy, programme, or actual projects before making the decision to proceed. Individual environmental topics are assessed and summarised in the Environmental Statement (ES) chapters.

This ES comprises three volumes which relate to each other as shown on the right side of this page, the descriptions of these documents should help you determine the most relevant documents for you.

The results of the assessment are available in the Environmental Statement and are summarised in this Non-Technical Summary (NTS).

The NTS is a document to the FS and is available for request as a paper copy by contacting Cottamstation@lancashire.gov.uk

Volumes of the ES

Volume 1 - Non Technical Summary This is a summary of the Cottam Parkway Railway Station Scheme and its likely significant effects on the environment. This document should be a useful place to start.

Volume 2 - Environment Statement This is the overall assessment of the environmental effects likely to be experienced during the construction and operational phases of the proposed Cottam Parkway Railway Station Scheme

Project Title: Cottam Parkway Railway Station ES Chapter Document Control **Document Reference:** ES-01-NTS-02 Chapter Title: Non-technical summary

Volume 3 - Appendices

This includes all the relevant technical documents and appendices associated with the Chapters covered in Volume 2.

Cottam Parkway Railway Station would be located in North West Preston, in approximately 9.2 hectares. The Preston to Blackpool railway line runs through the southern area of the site. To the north of the site is the Cottam Link Road with Lancaster Canal running centrally through the site from east to west. To the east is Lea Road and to the west is the Preston Western Distributor Road (PWDR).

The site lies within open countryside and is currently used for grazing cattle.



Why is the station needed?

North West Preston was identified in the Preston City Local Plan 2012 – 2026 as an area of strategic residential development. As the population in the region grows, it will be necessary to provide alternative and sustainable transport routes to the main employment centres of the region: Preston, Blackpool, Manchester and beyond. The station would provide sustainable transport infrastructure for the area of Ingol, Lea and Cottam. The benefits of the station would include:

Better rail access to support the delivery of new homes in North West Preston and provide rail service to a catchment area of around 12,000 homes in Cottam, Ingol and Lea.

- The new station would encourage a shift in journeys from road to rail. This would help to reduce traffic at key congestion hotspots on Preston's road network.
- There would be improved access via the Preston Western Distributor Road (scheduled to open in 2023) to existing and new employment opportunities at the Enterprise Zone at Warton.
- The station would provide more options for people to travel in and around the city and beyond.

Station Design

- The railway station forecourt would be located between two car parks and to the south of the forecourt would be the railway station building. The railway station forecourt would provide two bus stops, together with passenger drop off and a taxi collection bay.
- The railway station building would be located between the station forecourt and the western end of the northern platform. For convenience and ease of access, the finished floor level of the railway station building would closely align with the proposed platform level. The railway station building would have a green roof and solar panels. There will be a footbridge including a lift to connect the northern and southern platforms.
- A Secondary Means of Escape would be constructed from the southern platform towards Lea Road with an emergency vehicle access.
- Cycle parking would be provided and would be split between the railway station terrace area and by lockable cycle lockers to the south of the station forecourt.
- A segregated footway and cycleway along the existing Sidgreaves Lane connecting to Lea Road and the creation of a bus gate onto Lea Road.
- A new access road to the railway station which originates from the new Cottam Link Road Roundabout.
- The construction of a pond by the new roundabout to sustainably collect water runoff from the road.
- Construction of a new bridge over the Lancaster Canal.



Consideration of Alternatives

Five alternative locations were considered for the railway station. These were:

- **Option 1** Located partly below the future Preston Western Distributor Road bridge and the overhead power lines
- Option 2 Situated closest to Lea Road
- Option 2a Situated adjacent to Darkinson Lane/Sidgreaves Lane

- **Option 2b** Situated between option 2 and 2a and is an optimised version of these two options
- **Option 3** Located on a curved portion of the track in the centre of the area and situated to the west of Darkinson Lane

Option 2b was selected as the preferred location for the Cottam Parkway Railway Station.



Environmental Topics

Each chapter topic summary has been divided into the construction effects during the building of the Scheme and operational effects during the life span of the Scheme, when functioning as a railway station.

Air Quality

- It is predicted that the impacts on air quality during the construction of the Scheme would be minimal. Dust from vehicles and earthworks will be managed and kept to a minimum.
- The Scheme is not predicted to result in a significant change in the levels of air pollution (NO2, PM10 and PM2.5) in the local area, meaning that the risks to ecosystem health and human health caused by poor air quality are very low.



Ecology

- The Scheme is mostly located in an area of improved grassland used for grazing which is of low biological importance. However, habitats and species of nature conservation importance have been found in the wider area. These include scattered mature broad-leaved trees; hedgerows; common toad, slow worm; breeding birds; wintering birds; bats; hedgehog; brown hare and otter.
- The Scheme would incorporate tree planting, hedgerow planting and habitat creation for wildlife. The Scheme aims to achieve a minimum of 10% net increase in biodiversity (the quality and total number of plant and animal species) in the local area.

- Creating new habitats for ecology and biodiversity would result in beneficial effects for all species over the long term. Most species would remain largely unaffected by the Scheme, but there would be some limited local negative impacts to the brown hare.
- The brown hare is a protected species in the UK and they are commonly found in grassland habitats and at woodland edges. As North West Preston is developed for residential use, brown hare habitat is reduced. Mitigation for the loss of brown hare habitat is not possible.



Landscape

The Scheme is an area of land on the urban edge of Cottam which has a predominantly gentle topography, with slightly steeper slopes along water courses. There is 1 Biological Heritage Site, a Grade II listed Bridge, ponds, watercourses and well-established hedgerows and trees all within the site area. The Cottam Link Road, the Lancaster Canal and the Preston to Fylde junction to Blackpool North line also cross the site area. These all contribute to the landscape and the areas character.

It is acknowledged that the landscape will be significantly adversely affected during the construction of the Scheme due to the presence of plant and construction material. The impacts, whilst not completely resolved will be mitigated against through the following (but not limited to); the screening of construction areas, controlled movement of vehicles and controlled lighting.



There will also be an adverse impact during the operation of the Scheme due to the presence of the station building, footbridge, car park and access road. Tree and hedgerow mitigation planting would not yet be established and would have limited reduction in impacts at opening year. The Scheme would not result in any significant effects by operation summer year 15 due to sufficient vegetation growth, although the loss of pre-development open green space would be permanent.





Cultural Heritage

A total of 69 cultural heritage assets have been identified in the area around the railway station. This number is made up of 38 archaeological sites and 26 historic buildings or structures including 6 Grade II Listed Buildings, and 5 historic landscape types. During construction, 14 of the 69 cultural heritage assets would be impacted. However, these impacts are thought to be non-significant once mitigation has been completed. Mitigation measures include:

- Archaeological earthwork survey to Historic England Level 1;
- Careful removal of the Possible Railway Milestone (Asset 52), safe storage, and reinstatement after construction;
- Evaluation through archaeological investigation; and,
- Development of a targeted archaeological mitigation strategy during detailed design.

Noise and Vibration ->>>

During construction it is likely that even with the inclusion of noise mitigation measures and limited working hours, some negative noise impacts would still occur. These effects are likely to impact Quaker Lodge, The Shires Ashton Lodge and Railway Cottages. The impacts would be temporary whilst the Scheme is being built. No vibration effects are anticipated during the construction.

Once the Scheme is open, it is predicted that a number of significant beneficial effects in the short-term will occur, resulting in a noise decrease along Lea Road, which would experience reduced traffic. Overall, no significant changes to current noise levels are predicted from the operation of the railway station and no negative impacts are predicted from road traffic noise.

Soils, Geology and Hydrogeology

During construction of the Scheme, it is likely that there would be no significant impacts on human health, groundwater and surface water. It is likely that there would be a significant adverse impact on soil quality during construction of the Scheme. Primarily, this would be a consequence of:

- The permanent sealing of soils; and,
- stripping, storage and/or compaction.

During the operation of the Scheme, the assessment has determined that there would be no significant impacts on human health, surface water, groundwater, buildings and infrastructure or soil quality.



Making previously reusable soils unsuitable for reuse through excavation,



The Water Environment

Four aspects of the water environment were assessed to determine the impact of the Scheme on the water environment. These were:

- Surface water quality;
- Hydromorphology (Physical character and water content of water bodies);
- · Flood risk; and,
- Groundwater.

No impacts on surface water quality are anticipated during the construction and operation of the Scheme.

No impacts on hydromorphology are anticipated during the construction and operation of the Scheme once mitigation has taken place. Flood risk from river flooding, surface water, groundwater, and canals were assessed. The development would be located within Flood Zone 1 and is therefore, generally at low risk of flooding.

A new culvert would be designed to reroute the Central Watercourse beneath the new station and a surface water management system would ensure the Scheme is safe from flooding without increasing the risk of flooding elsewhere. Further groundwater assessment would take place at a later stage to fully assess the potential impacts of the Scheme.



Climate Change

Climate change resilience

Climate change resilience refers to the ability of the Scheme to adapt to the expected changes in climate over its operational lifetime.

In the short term, the effects from the construction of the Scheme are unlikely to cause any serious negative impacts on the

climate change resilience of the Scheme. This is due to the mitigation that would be implemented, the nature of the construction activities and the dates of the construction period.

During the operation of the Scheme, it is likely that there would be no significant impacts as a result of the Scheme in respect of climate change. The following mitigation measures are proposed to tackle climate change resilience of the Scheme:

 Consideration of the use of construction materials with superior properties (such as increased tolerance to fluctuating temperatures) to be included within detailed designs;

>>>

Consideration of climate change projections to be within maintenance plans;

•

٠

- Drainage systems designed to protect against a return period of 1-in-100 years flood event;
- Contingency plans for dealing with severe weather events; and,
- Regular maintenance of assets.

Human Health –

The potential impacts of the Scheme on human health were assessed.

Some minor to major beneficial impacts are predicted for the surrounding population once mitigation measures have been implemented and no significant adverse impacts on human health are predicted to arise. Resultingly, the Scheme would improve the health outcomes for the population of Ingol, Cottam and Lea. Health benefits of the station development include:

- Providing direct and indirect employment opportunities in the area;
- Providing sustainable transport options for work and leisure; and,
- Improving the walking and cycling infrastructure in the area which would also increase road safety for pedestrians.



Traffic and Transport

Following analysis, the traffic and transport assessment found that the Scheme would increase public transport accessibility for Cottam and wider North West Preston area, with areas such as Blackpool, South Ribble and Wigan all becoming easily accessible by public transport.

The Scheme is predicted to generate 1,146 daily passengers in 2024, increasing to 1,248 in 2039.

The station would require a Public Right of Way that currently crosses the site from west to east to be rerouted. Once rerouted, the impact on footpath users would be negligible.

Traffic modelling indicates that there would be negligible impact upon the capacity, safety or operation of the surrounding highway network as a result of the Scheme, therefore no mitigation measures would be required.



A slight beneficial operational impact on pedestrians and cyclists is predicted as wellconnected segregated cycling and walking infrastructure would be built.

A large beneficial effect on public transport would arise as a result of significantly improved access to the railway network, despite some bus route journey times being negatively impacted during the construction and operational phases.







Land use and Accessibility

Land use and Accessibility is the predicted impact that the Scheme would have on land use in the local area and the ability of users to access land, property, infrastructure, businesses and community facilities.

As construction begins on the Scheme, some disruption is expected to the local road network and a Public Rights of Way. Suitable diversions and traffic management would help to minimise the impact this may have.

During the operation of the Scheme, it is predicted that walkers, cyclists and horse riders in the area around the Scheme would benefit from the improved cycle and pedestrian infrastructure that would be constructed for the access.





However, public footpath would be diverted which may have an impact on journey times for footpath users.

Landowners and local businesses are predicted to experience a negligible beneficial impact associated with the improvements to the highway network allowing shortened journey times.



Materials and Waste

Building the railway station would require resources such as, concrete, tarmac, bricks and topsoil. This would result in a reduced availability of material resources such as aggregates.

The Scheme would also involve substantial earthworks.

The Scheme would generate other types of waste which require onsite waste management. The Scheme would rely on the local waste management facilities and would lead to the permanent reduction in landfill capacity. However, the independent assessment of the likely significant

environmental effects arising from waste materials or the use of natural resources has been scoped out by the Environmental Impact Assessment. These topic are also to be sufficiently addressed within the Scheme design, relevant discipline chapters and management plans.

Cumulative Impacts \rightarrow

Cumulative impacts are the impacts of the Scheme that may not be a cause for concern when assessed individually but may become more serious when looked at in the context of the wider area.

With North West Preston undertaking large amounts of residential building and transport infrastructure development, it is likely that cumulative effects may be present.

Without mitigation, the Scheme would contribute to the removal of hedgerows and trees that would further impact the local ecology by reducing habitat. However, the Scheme proposes to mitigate these losses by achieving a 10% increase in the total plants and

animal species present in the local area to compensate for the removal of hedgerows, trees and general loss of habitat.

The exception to this would be the loss of habitat for brown hare. Due to the amount of development in the area, brown hare have undergone a significant amount of habitat loss. The Scheme would contribute to the reduction of habitat for brown hare. Mitigation for this loss of habitat is not possible.





Summary

Throughout this document, it has been shown that there are relatively few significant effects anticipated as a result of the Scheme. When mitigating for the Scheme, we first choose to avoid any potential effects. If this is not possible then we would seek to minimise any potential adverse effects. Failing this, the next option would be to restore the impacted areas. Ultimately, if this is not possible, we would seek to offset any impacts that remain. This is known as the mitigation hierarchy. After mitigation, these significant effects are confined to the following EIA topics:

- Soils, Geology and Hydrology; and
- Ecology.

These and any other impacts likely to have an adverse impact on the environment would be mitigated through:

 The Construction Environmental Management Plan (CEMP). The CEMP would address working hours, traffic management measures, ecological mitigation, best on-site practice and measures to address potential pollution sources, waste management, noise, dust and vibration creation.

 An Environmental Masterplan, which incorporates landscaping and ecological mitigation, habitat creation, and sustainable drainage (Sustainable Drainage Systems). This Environmental Masterplan would seek to reduce impacts on the environment and where possible enhance the local surrounding area.

>>>

